Author’s response to reviews

Title: High-risk human papillomavirus (HPV) screening and detection in healthy patient saliva samples: a pilot study.

Authors:

Deidre O Turner (deidreoturner@yahoo.com)
Shelley J Williams-Cocks (scocks@medicine.nevada.edu)
Ryan Bullen (ryan.bullen@sdm.unlv.edu)
Jeremy Catmull (jeremy.catmull@sdm.unlv.edu)
Jesse Falk (jesse.falk@sdm.unlv.edu)
Danny Martin (daniel.martin@sdm.unlv.edu)
Jarom Mauer (jarom.mauer@sdm.unlv.edu)
Annabel E Barber (barber@medicine.nevada.edu)
Robert C Wang (rwang@medicine.nevada.edu)
Shawn L Gerstenberger (shawn.gerstenberger@unlv.edu)
Karl Kingsley (karl.kingsley@unlv.edu)

Version: 4 Date: 4 October 2011

Author's response to reviews:

All formatting changes have been made.

Major revisions: Abstract in submission system and manuscript are now identical; Authors' contributions has been revised; Accession numbers are now properly cited; Highlighting and tracking have been removed; All text is now in black; Email has been correctly formatted; Reference 61 has been updated; Figures have been cropped.

Many thanks for your assistance.